Author's response to reviews

Title: Mesenteric Panniculitis with pedal edema: A case report

Authors:

Abdul M Zafar (amueed@gmail.com)
Muhammad A Rauf (ameenrauf@gmail.com)
Tabish Chawla (tabish.chawla@aku.edu)
Gule Khanda (gule.khanda@aku.edu)

Version: 5 Date: 30 September 2008

Author's response to reviews: see over
Dear Editor,

Greetings,

The authors would like to thank the respected reviewers for their kind comments. These have been incorporated into the revised manuscript; details are presented below.

Kindly keep us informed of the review/acceptance status of the article.

Best Regards,

Dr. Abdul Mueed Zafar

Aga Khan University,

Karachi, Pakistan
Reviewer's report A

Reviewer: Nereo Vettoretto

Comments to authors: Good review of the report. It’s a pity not to have control CT scan or histology picture. I would explain why in the paper. It would not diminish the value of the reported case.

Authors’ response: The reason for omitting a repeat CT scan has been mentioned in Paragraph 4, line 3 of Case Presentation. Our health care system is privately funded and the patient’s monthly income was less than the cost of a single CT scan.

The histological evaluation was not done at our institution but at a military hospital. We were unable to obtain histology images due to bureaucratic hurdles. We suspect that the readers will not find this information pertinent.

Reviewer's report B

Reviewer: Angel Popkharitov

Comments to authors: I have no remarks regarding the revised article. The authors took into consideration my suggestions. I believe with these changes the paper could be accepted.

Authors’ response: The authors would like to express their gratitude for the reviewer’s time and constructive feedback.